Author's response to reviews

Title: Biomonitoring of bisphenol A concentrations in maternal and umbilical cord blood in regard to birth outcomes and adipokine expression: a birth cohort study in Taiwan

Authors:

Wei-Chun Chou (d9712871@oz.nthu.edu.tw)
Jyh-Larng Chen (jlchen@mail.ypu.edu.tw)
Chung-Fen Lin (chungfen1119@yahoo.com.tw)
Yi-Chun Chen (cyc@mail.cjcu.edu.tw)
Feng-Cheng Shih (fengcheng@mail.ypu.edu.tw)
Chun-Yu Chuang (cychuang@mx.nthu.edu.tw)

Version: 6 Date: 1 November 2011

Author's response to reviews:

Nov 1, 2011

MS No. 1312412857516153: Biomonitoring of bisphenol A concentrations in maternal and umbilical cord blood in regard to birth outcomes and adipokine expression: a birth cohort study in Taiwan

Dear Dr. Grandjean and Dr. Herz:

We have revised the manuscript according to the suggestions, and checked the typographical errors to make the final correction. In the following, we provide a point-by-point description of the changes made.

Tables: We include Tables in the manuscript in a tables section following the references.

References: In last revision, Reference 43 within the main text is a typographical error, and has been corrected to Reference 42.

Fig 1 refers to A and B: We describe Figure 1(A) and Figure 1(A) in the main text.

We revise to use “concentration” rather than “level”.

Figures: The Figure legend is in the manuscript file after the reference list. The figures are uploaded in “JPEG” format and numbered automatically.

Best regards,

Chun-Yu Chuang
Associate professor